Author’s response to reviews

Title: MR Imaging Features of Orbital Langerhans Cell Histiocytosis

Authors:

chunnan wu (wuchunnan_516@163.com)
xuetao Mu (muxuetao091@163.com)
yan Hei (heiyan842@163.com)
pengyu lan (shiyuen07@163.com)
kuncheng li (cjr.likuncheng@vip.163.com)

Version: 6 Date: 28 Oct 2019

Author’s response to reviews:

Response to the comments:
1. Consent for publication
In your ‘Consent for publication’ section of your declarations, please clearly state if the parents or guardians of the minor (aged under 18) study participants gave written consent for their personal or clinical details along with any identifying images to be published in this study. Alternatively, please anonymise the identifying information in the manuscript and supplementary files especially (e.g. grouping specific ages into age ranges - for instance the age "55" could be listed as "50-60").
Response:
In the revised manuscript, we have added the information.

2. Authors’ contributions
Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript in its current state.
Response:
In the revised manuscript, we have added the information.

3. Remove supplementary files
Please remove the supplementary files “Reviewer reports R3.doc” from the manuscript, as these files are not required to be attached to the manuscript at this stage of the editorial process.
Response:
In the revised manuscript, we have removed it.

4. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response:
In the revised manuscript, we have corrected it.